Author's response to reviews

Title: Neuroprotective effects of Eucommiae Ulmoides Oliv. and its bioactive constituent work via ameliorating the ubiquitin-proteasome system

Authors:

Hong Guo (cacti1983@163.com)
Fang Shi (sflrq@163.com)
Meijiao Li (meijiaojiao851020@163.com)
Qingqing Liu (liuqingqing8913@126.com)
Bin Yu (yubin_771115@hotmail.com)
Limin Hu (huliminth@126.com)

Version: 7 Date: 21 April 2015

Author's response to reviews: see over
Title: Neuroprotective effects of *Eucommiae Ulmoides* Oliv. and its bioactive constituent work via ameliorating the ubiquitin- proteasome system

Author: Hong Guo, Fang Shi, Meijiao Li, Qingqing Liu, Bin Yu, Limin Hu*
Tianjin University of Traditional Chinese Medicine, Tianjin State Key Laboratory of Modern Chinese Medicine, Tianjin Key Laboratory of Traditional Chinese Medicine Pharmacology, Tianjin 300193, China

*Author for correspondence*: Limin Hu, MD, professor. Institute of Traditional Chinese Medicine, Tianjin University of Traditional Chinese Medicine, #312 Anshan xi road, Nankai district, Tianjin 300193, China. E-mail: huliminth@126.com; Tel: (+86) 22-59596166; Fax: (+86) 22-59596168

Key words: *Eucommia ulmoides*; Parkinson’s disease; ubiquitin- proteasome system

Declaration:
This manuscript/data has not been submitted elsewhere for publication. All of the listed authors have read and approved the submitted manuscript. The present study was performed according to international, national and institutional rules.
Dear editor,

This is a revision of the manuscript submitted in BMC Complementary and Alternative Medicine entitled “Neuroprotective effects of *Eucommiae Ulmoides* Oliv. and its bioactive constituent work via ameliorating the ubiquitin-proteasome system” (ID: 1235502227142721). Thank you for your letter and for the reviewers’ comments concerning our manuscript. We apologize for our poor English writing in the manuscript and we have corrected grammatical/stylistic errors by a professional language editing service that you recommend. The changes were marked-up by the function of amendment in the revision.

To reviewer 2:

**See attached file for additional Minor Essential revisions to grammar.**

We apologize for our poor English writing in the manuscript and sincerely thank the reviewer’s corrections. We have modified the errors and marked the changes in the revision.

Again, we sincerely appreciate the time and effort by the Editor and Reviewer in the critique of this manuscript.

With best regards,

Hong Guo